Author's response to reviews

Title: Human papillomavirus infections in women seeking cervical Papanicolaou cytology of Durango, Mexico: prevalence and genotypes.

Authors:

Luis Francisco Sanchez Anguiano (lfsanguiano@hotmail.com)
Cosme Alvarado-Esquivel (alvaradocosme@yahoo.com)
Miguel Arturo Reyes-Romero (lfsanguiano@hotmail.com)
Margarita Carrera-Rodriguez (lfsanguiano@hotmail.com)

Version: 4 Date: 22 November 2005

Author’s response to reviews:

Dear Editor,

Thank you for your kind email.

We have added in the methods section of the manuscript the following details on the ethical approval:

Ethical aspects
This study was approved by the institutional ethical committee. The purpose and procedures of the study were explained to all participants, and a written informed consent was obtained from all of them.

Kindest regards,

Dr. Luis Francisco Sanchez-Anguiano.